Author's response to reviews

Title: Major depression, fibromyalgia and labour force participation: A population-based cross-sectional study

Authors:

Aliya Kassam (aliya.kassam@iop.kcl.ac.uk)
Scott B Patten (patten@ucalgary.ca)

Version: 3 Date: 19 December 2005

Author's response to reviews:

This resubmission involves format changes referenced in an email sent to me on December 19th, 2005:

1. Abstract checked - is identical to that in manuscript.
2. Authors names in manuscript changed (initial added), identical to manuscript.
3. Introduction renamed "background."
4. All underlining removed.
5. The web site previously reference in the text has been made into a reference.
6. The format of the websites in the acknowledgements has been corrected.
7. The order of the sections has been changed to conform with requirements.
8. The entire manuscript has been re-read for minor errors, one was found and corrected.

Scott Patten